
Authors:

Kathleen McLoughlin (kathleen.mcloughlin@nuim.ie)
Jim Rhatigan (jim.rhatigan@milfordcarecentre.ie)
Sinead McGilloway (sinead.mcgilloway@nuim.ie)
Allan Kellehear (a.kellehear@bradford.ac.uk)
Michael Lucey (m.lucey@milfordcarecentre.ie)
Feargal Twomey (f.twomey@milfordcarecentre.ie)
Marian Conroy (m.conroy@milfordcarecentre.ie)
Emilio Herrera-Molina (emilio.herrera@newhealthfoundation.org)
Suresh Kumar (drsuresh.kumar@gmail.com)
Mairead Furlong (mairead.furlong@nuim.ie)
Joanne Callinan (j.callinan@milfordcarecentre.ie)
Max Watson (alimaxuk@yahoo.com)
David Currow (david.currow@flinders.edu.au)
Christopher Bailey (C.D.Bailey@soton.ac.uk)

Version: 2
Date: 2 March 2015

Author’s response to reviews:

Please note ISRCTN number has now been confirmed and is included.